To what extent do disease severity and illness perceptions explain depression, anxiety and quality of life in hidradenitis suppurativa?*

A. Pavon Blanco (1),¹ M.A. Turner,^{1,2} G. Petrof (1)² and J. Weinman¹

¹King's College London, London WC2R 2LS, U.K.

²St John's Institute of Dermatology, Guy's and St Thomas' NHS Foundation Trust, London, U.K.

Summary

Correspondence

Alicia Pavon Blanco. E-mail: alicia.pavonblanco@yahoo.co.uk

Accepted for publication 19 August 2018

Funding sources

IMPARTS is funded partly by King's Health Partners and partly by the National Institute for Health Research (NIHR) Biomedical Research Centre and Dementia Unit at South London and Maudsley NHS Foundation Trust and King's College London. The research was also funded and supported by the NIHR Biomedical Research Centre based at Guy's and St Thomas' NHS Foundation Trust and King's College London. The views expressed are those of the authors and not necessarily those of the NHS, the NIHR or the Department of Health.

Conflicts of interest

M.A.T. received funding from AbbVie for his salary in 2014–2015 and has received honoraria for advisory panel work from Lilly and for workshops from Novartis.

*Plain language summary available online

DOI 10.1111/bjd.17123

Background Hidradenitis suppurativa (HS) can have significant psychological consequences and affect quality of life (QoL). This has been associated with disease severity. However, it has not been established whether these effects are more strongly related to the severity of the disease, as rated by the clinician, or to the patient's perception of their condition.

Objectives To examine the relationships between disease severity and illness perceptions, and depression, anxiety and QoL in HS.

Methods This study was cross-sectional in design. In total, 211 patients with HS completed the Brief Illness Perception Questionnaire (BIPQ), the Patient's Health Questionnaire-2 (PHQ-2), the Generalized Anxiety Disorder-2 (GAD-2) and the Dermatology Life Quality Index (DLQI). HS severity was assessed by the clinician, using the Hurley staging system.

Results Patients with HS perceived their condition as chronic – having many symptoms, severe consequences and a negative emotional influence – and felt low personal control over their illness. Self-reports showed significant levels of depression, anxiety and impaired QoL, which were strongly associated with illness perceptions. Hierarchical regression analyses revealed that illness perceptions explained a much greater proportion of variance in depression, anxiety and QoL than the traditional explanatory variable, disease severity.

Conclusions HS can severely impair psychological well-being and QoL, which are more strongly associated with the person's beliefs about their illness than clinicians' severity assessments. Therefore, illness perceptions may be useful in the routine assessment of patients with HS and may provide a strong basis for interventions aimed at improving their psychological well-being and QoL.

What's already known about this topic?

- Hidradenitis suppurativa (HS) is known to be a highly distressing and debilitating skin condition that has a profound effect on quality of life (QoL), mood and anxiety.
- It has been assumed that these effects are directly related to disease severity.
- No previous studies have investigated patients' perceptions of their HS, and whether these are related to disease severity and/or their QoL and mood.

What does this study add?

• This study examines the illness perceptions of patients with HS, and the extent to which these are related to clinician-rated disease severity, mood and QoL.

What are the clinical implications of this work?

- The results suggest that illness perceptions are a stronger indicator of which patients with HS are most likely to be depressed, anxious or impaired in their QoL, relative to this skin disease's clinical measurement.
- There is certainly scope for more routine assessment of patients' perceptions and for the development of interventions aimed at targeting negative illness beliefs with the aim of improving outcomes.

Hidradenitis suppurativa (HS), also called acne inversa, is a long-term, recurrent, debilitating inflammatory dermatological disease of the hair follicles. It is characterized by abscesses, sinus tracts and scarring, mainly affecting the apocrine-gland-bearing regions (axillae and anogenital and inframammary areas).^{1,2} Persistent lesions may result in sinus tracts, usually accompanied by pain and malodorous suppuration.^{1,2} Treatment varies according to disease severity, which is usually assessed using the Hurley³ staging system.¹ HS is seen more among women. It tends to develop in patients in their 20s and its prevalence declines in people aged over 50 years.^{4,5} Prevalence estimates of 1–2% have been reported in Europe.^{5–}

HS can have an extremely negative impact on patients' psychosocial and daily functioning and has been associated with anxiety, depression and impaired quality of life (QoL).^{8–13} The occurrence of lesions in exposed areas and intimate sites, as well as odorous discharge that can stain clothing,^{2,14} can result in difficulties with self-image and feelings such as stigma and shame for people with HS. Although studies of the impact of HS on psychological well-being and QoL are very limited, a higher prevalence of depression and poorer QoL have been found for HS compared with other dermatoses such as psoriasis, acne and chronic urticaria.^{8–11,13} Depression and impaired QoL have also been linked to HS severity according to the Hurley staging system^{8,9,13} and the Sartorius¹⁵ scoring system,¹³ although one study⁹ found the association between depression and Hurley stage weak and borderline significant.

Increasing evidence among other chronic illnesses indicates that the individual's subjective beliefs about their condition are strongly related to health outcomes including depression, anxiety, QoL and recovery.^{16–22} These beliefs, which are collectively referred to as illness perceptions, comprise the patient's own ideas about the nature, cause, controllability, time course and consequences of their condition.²³ Compared with clinical assessments of disease severity and demographics, illness perceptions have been found to be more strongly related to depression, anxiety and QoL in alopecia,^{17,22} heart failure¹⁸ and multiple sclerosis.²⁴ Thus, patients' illness perceptions may contribute to health outcomes more significantly than HS severity.

To date, no research has attempted to understand the nature of the illness perceptions of patients with HS and how they impact on health-related outcomes, such as depression, anxiety and QoL. Previous studies have focused mainly on describing clinical aspects of the disease and their relationships with health outcomes (depression, QoL and functioning)^{8,9} and lifestyle behaviours (smoking and obesity).^{5,25} Therefore, the aims of the present study were to explore firstly illness perceptions in patients with HS and secondly their relationships with health outcomes (depression, anxiety and QoL) independently of HS clinical severity. Thus, we hoped to investigate whether patient outcomes are more strongly related to clinicians' ratings of disease severity or to the patient's perception of their condition.

Patients and methods

This was an exploratory cross-sectional study. Data were collected between July 2014 and January 2015 at an HS tertiary clinic of a large teaching hospital in London. Given that the self-reported data were routinely collected as part of the Integrating Mental and Physical Healthcare: Research Training and Services (IMPARTS) initiative, which links electronic patient records, patients did not formally consent to participate but were informed that they could opt out of their data being used for research purposes. This conforms with the principles of the IMPARTS programme ethics approval (IMPARTS Research Database REC reference: 12/SC/0422).

Patients and procedure

All consecutive patients attending the HS tertiary service were invited to complete a series of self-reported questionnaires on a tablet device connected to the electronic patient record software via the hospital's Wi-Fi. Registered nurses trained in using the tablet informed patients about the data collection procedure, followed by an opt-out consent process. Patients visiting the clinic on more than one occasion throughout the data collection period were included during their first visit only. We excluded those who had visited the clinic between July 2014 and January 2015 who did not have HS (n = 10), thus an initial cohort of 221 participants was reduced to the final sample of 211.

Psychological and quality-of-life measures

Brief Illness Perception Questionnaire

The Brief Illness Perception Questionnaire (BIPQ)²⁶ is a validated, self-reported and reliable nine-item scale evaluating cognitive and emotional representations of illness. Consequences, assesses the patient's perceived health outcomes of living with the illness; timeline, the perception of the duration of the disease; personal control and treatment control, the beliefs about the controllability of the disease; identity, the disease label and associated symptoms; concern and emotional response, emotional reactions to the illness; understanding, the patient's comprehensibility of their condition; and causes, the perceived aetiology of the disease (not included in this study). Each item is rated on a 10-point scale, where higher scores represent stronger endorsement with the illness perception. As recommended by the scale's authors,²⁷ each individual item value, not the overall score, was considered in this investigation.

Patient Health Questionnaire-2 and Generalized Anxiety Disorder-2

The Patient Health Questionnaire-2 $(PHQ-2)^{28}$ is a validated brief self-reported screening measure for assessing the frequency of depression and anhedonia within the last 2 weeks. It has been shown to be reliable in screening for depression in long-term conditions such as lung cancer.²⁹

The Generalized Anxiety Disorder-2 scale $(GAD-2)^{30}$ is a reliable and standardized self-reported instrument for screening for generalized anxiety disorder in the last 2 weeks. It has been shown to be reliable in chronic conditions such as cardiovascular disease.³¹

For PHQ-2 and GAD-2, each item is scored on a four-point rating scale varying from 0 (not at all) to 3 (nearly every day), giving a score range of 0–6. Higher scores represent more severe symptoms, with a recommended optimal cut-off point of 3 reflecting core signs of depression disorder²⁸ or generalized anxiety disorder.³⁰ The Cronbach's alpha value for GAD-2 in the current sample was 0-89.

Dermatology Life Quality Index

The Dermatology Life Quality Index $(DLQI)^{32}$ is a 10-item validated and widely used self-reported measure to assess QoL in dermatological illnesses. The full score range is 0–30, with higher scores indicating more impaired QoL. Interpretations of cut-off points are 0–1, no effect on the patient's life; 2–5, small effect; 6–10, moderate effect; 11–20, very large effect and 21–30, extremely large effect.³³ Its internal consistency value in the current sample was 0.90.

Clinical and demographic information

The clinical stage of the disease and demographic information of the included patients were extracted from the electronic patient record. The Hurley staging system,³ comprising three clinical stages – stage I (mild), stage II (moderate) and stage III (severe) – was used by clinicians to assess the severity of HS disease.

Statistical analyses

Statistical analyses were performed using SPSS for Windows version 22.0 (IBM, Armonk, NY, U.S.A.).³⁴ Post hoc power analysis on G*power $3 \cdot 1^{35}$ using the n = 211 sample size indicated 98% and 99% statistical power at a 0.05 alpha level (two-tailed) for detecting a medium and large effect size, respectively. The five-stage classification of Hurley's severity used for this sample was clustered into stages I, II and III following the original staging system and to allow for comparison of the results with the literature (Table 1).

There were no missing values, and preliminary analysis showed that nonparametric tests were appropriate for the data. Thus, Spearman's correlation coefficients were used to explore the associations between illness perceptions and age. Mann– Whitney U-tests were used to examine significant differences between illness perceptions across genders (data were recorded as 'gender'). We used Jonckheere–Terpstra tests to examine for ordered statistical differences in illness perceptions among the three disease severity groups, followed by post hoc pairwise comparisons. Hierarchical multiple regression models examined the contribution of illness perceptions to depression, anxiety and QoL, independently of HS clinical stage.

To control for the possible influences of confounders on the outcome variables, demographics (age and gender) and Hurley severity stage were entered into the regression analyses at steps 1 and 2, respectively. The BIPQ items were entered at step 3. The bias-corrected accelerated bootstrapping method (1000 samples) was applied to correct for heteroscedasticity and potential influential points and outliers. Statistical significance was defined as P < 0.05 given the exploratory nature of the present study.

Results

Baseline characteristics

The study included a total of 211 patients with HS aged 17-71 years (mean \pm SD, $38\cdot2 \pm 11\cdot8$), with the majority being female (n = 126, 60%). The sociodemographic and clinical characteristics of the study cohort are displayed in Table 1.

 Table 1
 Sociodemographic and clinical characteristics of the study

 sample with hidradenitis suppurativa (HS)

Gender, n (%)	
Female	126 (60)
Male	85 (40)
Age (years), mean \pm SD	$38\cdot2 \pm 11\cdot8$
HS Hurley stage severity, n (%)	
Stage Iª	45 (21)
Stage II ^b	64 (30)
Stage III	109 (52)
Current smoker, n/N (%)	86/210 (41)
Recreational drug user, n/N (%)	29/202 (14)

 $^{\mathrm{a}}\mathrm{Clustered}$ from stages I and I/II. $^{\mathrm{b}}\mathrm{Clustered}$ from stages II and II/III.

 Table 2 Distribution of Brief Illness Perception Questionnaire (BIPQ)

 scores and number and percentage of participants scoring above the

 scale midpoint

Dimension	Mean	SD	Median	Percentage of participants above midpoint (> 5)
Consequences	6.97	2.67	7	76
Timeline	8.49	2.49	10	84
Personal control	3.03	3.11	2	23
Treatment control	5.30	3.02	5	46
Identity	7.17	2.42	7	76
Concern	8.23	2.21	9	86
Understanding	7.03	2.62	7	72
Emotional response	7.26	2.92	8	78

Higher scores represent stronger endorsement of items.

Illness perceptions

Table 2 illustrates the patients' BIPQ scores and the number and percentage of patients scoring above the midpoint (> 5) for each illness perception. Most patients perceived HS as chronic and were worried about their illness. Most patients also felt that HS affects their life greatly (consequences) and that they had little control over their disease, experienced high levels of HS symptoms (identity) and had a strongly adverse emotional response to HS. They perceived their illness as fairly controllable through treatment and had a good understanding of HS.

Self-reported depression, anxiety and quality of life

Typically, participants' scores were below the threshold for clinically significant symptoms of depression (PHQ-2: median 2, mean $2 \cdot 12 \pm 1 \cdot 88$) and anxiety (GAD-2: median 2, mean $1 \cdot 88 \pm 1 \cdot 86$). However, around one-third (73 of 211, 35%) and around one-quarter (56 of 211, 27%) of participants scored equal to or above the threshold representing core symptoms of depression and anxiety, respectively.^{28,30} Generally, participants reported very large effects of HS symptoms on QoL (DLQI: median 15, mean $14 \cdot 81 \pm 8 \cdot 45$). Furthermore, over one-third (81 of 211, 38%) recorded a DLQI score between 11 and 20, representing a very large impact of HS on QoL, while almost one-third (59 of 211, 28%) scored in the range 21–30, indicating an extremely large effect.³³

Variations in illness perceptions by sociodemographics and hidradenitis suppurativa clinical severity

Spearman's correlations, Mann–Whitney U-tests and Kruskal– Wallis Dunn–Bonferroni tests were conducted to determine the differences in illness perceptions according to age, gender and disease severity (Tables 3, 4 and 5, respectively). Spearman's coefficients revealed that with increasing age, patients Table 3 Spearman's rho correlation coefficients between illness perceptions and age, depression (PHQ-2), anxiety (GAD-2) and quality of life (DLQI)

	Age	PHQ-2 GAD-2		DLQI				
Consequences	-0.18**	0.52**	0.48**	0.65**				
Timeline	0.03	0.16*	0.07	0.11				
Personal control	0.01	-0.17*	-0.19**	-0.21**				
Treatment control	0.07	-0.20**	-0.16*	-0.20**				
Identity	-0.09	0.39**	0.29**	0.59**				
Concern	-0.15*	0.29**	0.23**	0.32**				
Understanding	-0.12	0.02	-0.02	0.04				
Emotional response	-0.26**	0.45**	0.52**	0.51**				
PHQ, Patient's Health Questionnaire; GAD, Generalized Anxiety								

Disorder; DLQI, Dermatology Life Quality Index. *P < 0.05, **P < 0.01 (two-tailed).

attributed significantly fewer adverse consequences to HS ($r_s = -0.18$, P = 0.008) and had a significantly lower emotional response to the illness ($r_s = -0.26$, P < 0.001). Mann– Whitney U-tests showed that perceptions of timeline and treatment control were significantly higher for women (medians 10 and 6, respectively) than for men (medians 10 and 5, respectively): U = 4374.5, z = -2.63, P = 0.009, r = -0.18and U = 4474.5, z = -2.04, P = 0.041, r = -0.14, respectively. Jonckheere–Terpstra tests demonstrated a significant trend of higher understanding of HS with higher disease severity (J = 8748, z = 3.76, P < 0.001). Pairwise comparisons revealed a significantly higher understanding for patients of Hurley stage III (median 8) vs. stage I (median 6); adjusted P < 0.001. There were no other significant differences between groups.

Relationships between illness perceptions and depression, anxiety and quality of life

Spearman's correlations were used to assess the relationships between illness perceptions and depression, anxiety and QoL, which are presented in Table 3. Overall, the results showed that higher levels of depression and anxiety and more impaired QoL were associated with more negative illness beliefs.

Three hierarchical multiple regressions further examined these associations (Tables S1–S3; see Supporting Information). The change in \mathbb{R}^2 ($\Delta \mathbb{R}^2$) was tested for significance to examine whether illness perceptions account for additional variance in depression, anxiety and QoL over and above that accounted for by disease severity and demographics.

For depression, disease severity accounted for a significant amount of variance over and above demographic variables: $\Delta R^2 = 0.03$, $\Delta F(2, 206) = 3.75$, P = 0.025. Entry of illness beliefs into the model produced the greatest and most significant increase, explaining an additional 30% of the variance in depression scores: $\Delta R^2 = 0.30$, $\Delta F(8, 198) = 11.52$, P < 0.001. Inspection of the beta-coefficients revealed that

Table 4	Variations	in	illness	perceptions	by	gende
---------	------------	----	---------	-------------	----	-------

	Male			Female			P_value	Effect size ^b	
Percentile	25th	25th 50th ^a		25th	50th ^a	75th	1 funde	Effect Size	
Consequences	6	7	10	6	7	9.25	0.84	-0.01	
Timeline	6	10	10	9	10	10	0.009	-0.18	
Personal control	0	3	5	0	2	5	0.60	-0.04	
Treatment control	4	6	8	3	5	7	0.041	-0.14	
Identity	6	7	10	6	7.50	9	0.70	-0.03	
Concern	7	10	10	7	9	10	0.26	-0.08	
Understanding	5	7	9.50	5	7	10	0.62	-0.03	
Emotional response	5	8	10	6	8	10	0.59	-0.04	

Higher scores signify stronger endorsement of the items. ^aMedian. ^bCalculation for effect size estimate: $r = Z/\sqrt{N}$, where Z is the z-score of the U-test statistic and N is the sample size.⁵³

Table 5 Variations in illness perceptions related to disease severity

	Hurley stage I			Hurley stage II			Hurley stage III			Jonckheere–Terpstra	
Percentile ^a	25th	50th	75th	25th	50th	75th	25th	50th	75th	J-value	P-value
Consequences	5	7	8	6	7	9	5	8	10	7626.5	0.18
Timeline	6	10	10	8	10	10	8	10	10	7413	0.31
Personal control	0	3	5.5	0	2	5	0	2.5	5	6973	0.96
Treatment control	4.5	5	7	2	5	7	3	6	8	7456.5	0.33
Identity	6	7	8	6	7	8.75	5	8	10	7599	0.20
Concern	7	9	10	7	8.5	10	7	10	10	7536	0.23
Understanding ^b	4.5	6	8	5	7	9	6	8	10	8748	< 0.001
Emotional response	6	8	10	6	8	10	5	8	10	6992	0.99

Higher illness perception scores signify stronger endorsement of the items. ^aThe 50th percentile is the median. ^bPost hoc tests as follows. Standard test statistics: stage I vs. II, 1.78; I vs. III, 3.62; II vs. III, 2.09. Adjusted P-values: stage I vs. II, 0.11; I vs. III, < 0.001; II vs. III, 0.055.

perceptions of severe consequences [$\beta = 0.36$, P = 0.001, 95% confidence interval (CI) 0.13–0.39] and lower treatment control ($\beta = -0.20$, P = 0.004, 95% CI -0.21 to -0.04) were both significant contributors to depressive symptoms.

Interestingly, entry of severity stage did not explain additional variance in anxiety over and above demographic variables: $\Delta R^2 = 0.01$, $\Delta F(2, 206) = 1.22$, P = 0.30. The addition of illness perceptions into the model again contributed to the highest and most significant increment (27%) in the explanatory power: $\Delta R^2 = 0.27$, $\Delta F(8, 198) = 9.59$, P < 0.001. A higher emotional response ($\beta = 0.36$, P = 0.001, 95% CI 0.14–0.32), more negative consequences ($\beta = 0.33$, P = 0.002, 95% CI 0.11–0.34) and lower concern about HS ($\beta = -0.16$, P = 0.027, 95% CI -0.28 to -0.00) were significantly associated with increased anxiety.

For QoL, disease severity added a significant amount of variance beyond that accounted for by demographics: $\Delta R^2 = 0.08$, $\Delta F(2, 206) = 9.45$, P < 0.001. The addition of illness perceptions made the greatest significant contribution, explaining 47% of the variance in QoL scores above that already explained by the previous variables: $\Delta R^2 = 0.47$, ΔF (8, 198) = 26.78, P < 0.001. Beliefs of greater adverse

consequences ($\beta = 0.41$, P = 0.001, 95% CI 0.94–1.66), lower treatment control ($\beta = -0.13$, P = 0.010, 95% CI -0.65 to -0.12), a stronger identity ($\beta = 0.29$, P = 0.001, 95% CI 0.61–1.38) and a larger emotional response ($\beta = 0.18$, P = 0.005, 95% CI 0.14–0.91) showed significant effects in the variance explained for impaired QoL.

Discussion

To the best of our knowledge, this is the first study to identify an HS illness perceptions profile and to explore the relationships between illness perceptions and health outcomes in this patient group. The results are consistent with previous findings about the high levels of depression, anxiety and impaired QoL found in individuals with HS.^{8–12} Interestingly, despite considerable variations in illness perceptions within the sample, our HS cohort showed higher perceptions of consequences, symptoms, concern and emotional response and lower personal and treatment control than patients with diabetes, asthma and myocardial infarction.^{26,36,37} Overall, there was a lack of associations between a person's illness beliefs and their disease severity stage, which aligns with past evidence in psoriasis²¹ and systemic lupus erythematosus.³⁸ This is an important finding as it shows that illness perceptions are not a function of disease severity in HS, and that it is possible to examine the extent to which outcomes, such as depression and QoL, are more strongly related to clinician-dependent (Hurley stage) or patient-dependent (BIPQ) severity assessments. This highlights the potential discordance between patients' and doctors' evaluations of HS also found in other conditions^{39,40} and discourages assumptions about how the patient sees their disease according to clinical measures.

In line with past research in alopecia^{17,22} and other longterm conditions,^{18,20–22,24} more negative illness beliefs were significantly intercorrelated with higher depression and anxiety and impaired QoL scores. Furthermore, after controlling for HS severity and demographics (age and gender) in multivariate analyses, illness perceptions explained more variance than disease severity and demographic factors together, for depression, anxiety and QoL. These findings are consistent with previous studies across a range of chronic conditions^{17,18,20,24,41} and suggest a greater importance of individual's beliefs about their HS than disease severity and demographics in accounting for health outcomes. Contrarily to our results, Matusiak et al.⁸ found that disease severity (measured by Hurley stage) was the most important factor influencing depression and QoL. Moreover, Kurek et al.¹³ demonstrated that the higher the disease severity (assessed by the Sartorius scale), the higher the risk of depression. However, neither of the previous studies measured illness perceptions.

The most consistent significant contributor to depression, anxiety and impaired QoL was negative consequences, which have been repeatedly documented as a strong predictor of health outcomes.^{17,20,21,24,27,41,42} BIPQ emotional response was significantly associated with anxiety and impaired QoL, whereas treatment control explained significant variance in depression and impaired QoL, which is consistent with other studies.^{17,27} Surprisingly, concern beliefs showed a different trend in the regression analyses, with higher concerns significantly contributing to a reduction in anxiety levels. This relationship may have been indirectly influenced by a third variable, which should be investigated in future studies.

These findings have important clinical implications. Practitioners have the potential to make a positive contribution to patients' psychological well-being and QoL by routinely assessing their illness perceptions rather than focusing solely on the clinical symptoms of their skin condition. Illness perceptions have been shown to be modifiable by brief psychoeducational programmes in other long-term conditions, with subsequent improvement in patient outcomes including depression, anxiety, QoL and disease symptoms.43-46 Given the higher incidence of depression and the lower QoL found in patients with HS compared with other dermatoses,^{8,9,11} it is of great importance to help people with HS deal with their unhelpful illness beliefs, such as perceived lack of treatment control over HS, perceived negative consequences, and their emotional reactions to the disease. When designing interventions for HS, it is crucial to take into account the considerable variation in illness perceptions found in the present study, as it has been previously. $^{\rm 37}$

We acknowledge that the study's cross-sectional design does not provide evidence of causality but only reveals associations among the variables. Prospective longitudinal studies can help examine patients' variability in illness perceptions and how they influence health outcomes over time. Additionally, participants were recruited from a single teaching hospital in London and therefore the sample might not represent the whole HS spectrum, compromising the generalizability of the findings. Future research should include patient beliefs about what causes HS, as it is one of the most modifiable illness perceptions.²⁷ Studies could also build on the results of this study and examine coping mechanisms and resilience in patients with HS along with how they might affect health outcomes, which has been touched upon previously.47,48 Finally, it will be interesting to see whether the results are replicated when employing dynamic HS scoring systems such as the International Hidradenitis Suppurativa Severity Score System,⁴⁹ and how illness perceptions relate to other validated and skin-specific subjective measures of disease burden such as PRISM (Pictorial Representation of Illness and Self Measure).^{50–52}

In conclusion, illness perceptions in patients with HS, rather than disease severity, showed consistently greater and more significant associations with depression, anxiety and QoL. This suggests that health professionals should target negative perceptions and emotions about the disease in treatment programmes. Further research is needed to assess the effectiveness of interventions aimed at altering unhelpful beliefs in patients with HS and their benefits in managing their condition.

Acknowledgments

We are very grateful to Dr Nemesha Desai, consultant dermatologist and clinical lead of the HS clinic at St John's Institute of Dermatology, for her clinical contributions; Dr Sam Norton, statistician and lecturer at King's College London, for his guidance with the statistical analyses; IMPARTS for allowing us to use their records of patients with HS; and especially to Anna Simpson for her assistance with the HS database. Most importantly, we wish to thank the people with HS who allowed their data to be used for this study and the dermatology consultants and nurses who supported this project and helped this patient group to complete the measures employed.

References

- Dhaou BB, Boussema F, Aydi Z et al. Hidradenitis suppurativa (Verneuil's disease). J Saudi Soc Dermatol Dermatol Surg 2013; 17:1–5.
- 2 Esmann S, Jemec GB. Psychosocial impact of hidradenitis suppurativa: a qualitative study. Acta Derm Venereol 2011; **91**:328–32.
- 3 Hurley H. Axillary hyperhidrosis, apocrine bromhidrosis, hidradenitis suppurativa, and familial benign pemphigus: surgical approach. In: Dermatological Surgery (Roenigk RK, Roenigk HH Jr, eds). New York: Marcel Dekker, 1989; 729–39.

- 4 Jemec GB, Heidenheim M, Nielsen NH. The prevalence of hidradenitis suppurativa and its potential precursor lesions. J Am Acad Dermatol 1996; 35:191-4.
- 5 Revuz JE, Canoui-Poitrine F, Wolkenstein P et al. Prevalence and factors associated with hidradenitis suppurativa: results from two case–control studies. J Am Acad Dermatol 2008; **59**:596–601.
- 6 Alikhan A, Lynch PJ, Eisen DB. Hidradenitis suppurativa: a comprehensive review. J Am Acad Dermatol 2009; 60:539–61.
- 7 Vinding G, Miller I, Zarchi K et al. The prevalence of inverse recurrent suppuration: a population-based study of possible hidradenitis suppurativa. Br J Dermatol 2014; 170:884–9.
- 8 Matusiak Ł, Bieniek A, Szepietowski JC. Psychophysical aspects of hidradenitis suppurativa. Acta Derm Venereol 2010; 90:264–8.
- 9 Onderdijk A, Van der Zee H, Esmann S et al. Depression in patients with hidradenitis suppurativa. J Eur Acad Dermatol Venereol 2013; 27:473–8.
- 10 Vazquez BG, Alikhan A, Weaver AL et al. Incidence of hidradenitis suppurativa and associated factors: a population-based study of Olmsted County, Minnesota. J Invest Dermatol 2013; 133:97–103.
- 11 von der Werth J, Jemec G. Morbidity in patients with hidradenitis suppurativa. Br J Dermatol 2001; 144:809–13.
- 12 Shavit E, Dreiher J, Freud T et al. Psychiatric comorbidities in 3207 patients with hidradenitis suppurativa. J Eur Acad Dermatol Venereol 2015; 29:371–6.
- 13 Kurek A, Peters J, Milena E et al. Depression is a frequent co-morbidity in patients with acne inversa. J Dtsch Dermatol Ges 2013; 11:743–9.
- 14 Ingram JR, Abbott R, Ghazavi M et al. The hidradenitis suppurativa priority setting partnership. Br J Dermatol 2014; **171**:1422–7.
- 15 Sartorius K, Lapins J, Emtestam L et al. Suggestions for uniform outcome variables when reporting treatment effects in hidradenitis suppurativa. Br J Dermatol 2003; 149:211–13.
- 16 Broadbent E, Ellis CJ, Gamble G, Petrie KJ. Changes in patient drawings of the heart identify slow recovery after myocardial infarction. Psychosom Med 2006; 68:910–13.
- 17 Cartwright T, Endean N, Porter A. Illness perceptions, coping and quality of life in patients with alopecia. Br J Dermatol 2009; 160:1034–9.
- 18 Morgan K, Villiers-Tuthill A, Barker M et al. The contribution of illness perception to psychological distress in heart failure patients. BMC Psychol 2014; 2:50.
- 19 Reynolds L, Broadbent E, Ellis CJ et al. Patients' drawings illustrate psychological and functional status in heart failure. J Psychosom Res 2007; 63:525–32.
- 20 Scharloo M, Kaptein A, Weinman J et al. Illness perceptions, coping and functioning in patients with rheumatoid arthritis, chronic obstructive pulmonary disease and psoriasis. J Psychosom Res 1998; 44:573-85.
- 21 Scharloo M, Kaptein A, Weinman J et al. Patients' illness perceptions and coping as predictors of functional status in psoriasis: a 1-year follow-up. Br J Dematol 2000; 142:899–907.
- 22 Chiang Y, Bundy C, Griffiths C et al. The role of beliefs: lessons from a pilot study on illness perception, psychological distress and quality of life in patients with primary cicatricial alopecia. Br J Dermatol 2015; **172**:130–7.
- 23 Petrie K, Weinman J. Why illness perceptions matter. Clin Med 2006; 6:536–9.
- 24 Jopson NM, Moss-Morris R. The role of illness severity and illness representations in adjusting to multiple sclerosis. J Psychosom Res 2003; 54:503–11.
- 25 Sartorius K, Emtestam L, Jemec G et al. Objective scoring of hidradenitis suppurativa reflecting the role of tobacco smoking and obesity. Br J Dermatol 2009; 161:831–9.

- 26 Broadbent E, Petrie KJ, Main J, Weinman J. The Brief Illness Perception Questionnaire. J Psychosom Res 2006; 60:631–7.
- 27 Broadbent E, Wilkes C, Koschwanez H et al. A systematic review and meta-analysis of the Brief Illness Perception Questionnaire. Psychol Health 2015; 30:1361–85.
- 28 Kroenke K, Spitzer RL, Williams JB. The Patient Health Questionnaire-2: validity of a two-item depression screener. Med Care 2003; 41:1284–92.
- 29 Randall JM, Voth R, Burnett E et al. Clinic-based depression screening in lung cancer patients using the PHQ-2 and PHQ-9 depression questionnaires: a pilot study. Support Care Cancer 2013; 21:1503-7.
- 30 Kroenke K, Spitzer RL, Williams JB et al. Anxiety disorders in primary care: prevalence, impairment, comorbidity, and detection. *Ann Intern Med* 2007; **146**:317–25.
- 31 Celano CM, Suarez L, Mastromauro C et al. Feasibility and utility of screening for depression and anxiety disorders in patients with cardiovascular disease. Circ Cardiovasc Qual Outcomes 2013; 6:498– 504.
- 32 Finlay AY, Khan G. Dermatology Life Quality Index (DLQI) a simple practical measure for routine clinical use. Clin Exp Dermatol 1994; 19:210–16.
- 33 Hongbo Y, Thomas CL, Harrison MA et al. Translating the science of quality of life into practice: what do dermatology life quality index scores mean? J Invest Dermatol 2005; 125:659-64.
- 34 IBM Corp. Released 2013. IBM SPSS Statistics for Windows, Version 22.0. Armonk, NY: IBM Corp.
- 35 Faul F, Erdfelder E, Lang A-G et al. G* Power 3: a flexible statistical power analysis program for the social, behavioral, and biomedical sciences. *Behav Res Methods* 2007; **39**:175–91.
- 36 Riverón GB, Guzmán MO, Miranda AL et al. Validation of the Brief Illness Perception Questionnaire (BIPQ) in hypertensive patients. Rev Psicol Trujillo (Peru) 2015; 15:78–91 (in Spanish).
- 37 Petrie KJ, Perry K, Broadbent E et al. A text message programme designed to modify patients' illness and treatment beliefs improves self-reported adherence to asthma preventer medication. Br J Health Psychol 2012; **17**:74–84.
- 38 Daleboudt GM, Broadbent E, Berger SP et al. Illness perceptions in patients with systemic lupus erythematosus and proliferative lupus nephritis. Lupus 2011; 20:290–8.
- 39 Chen P, Broadbent E, Coomarasamy C et al. Illness perception in association with psychological functioning in patients with discoid lupus erythematosus. Br J Dermatol 2015; 173:824–6.
- 40 Jones-Caballero M, Chren M, Soler B et al. Quality of life in mild to moderate acne: relationship to clinical severity and factors influencing change with treatment. J Eur Acad Dermatol Venereol 2007; 21:219–26.
- 41 Moss-Morris R, Petrie KJ, Weinman J. Functioning in chronic fatigue syndrome: do illness perceptions play a regulatory role? Br J Health Psychol 1996; 1:15–25.
- 42 Hagger MS, Orbell S. A meta-analytic review of the common-sense model of illness representations. Psychol Health 2003; 18:141–84.
- 43 Broadbent E, Ellis CJ, Thomas J et al. Further development of an illness perception intervention for myocardial infarction patients: a randomized controlled trial. J Psychosomat Res 2009; 67:17–23.
- 44 Hunter MS, Coventry S, Hamed H et al. Evaluation of a group cognitive behavioural intervention for women suffering from menopausal symptoms following breast cancer treatment. Psychooncology 2009; 18:560–3.
- 45 Kaptein AA, Bijsterbosch J, Scharloo M et al. Using the common sense model of illness perceptions to examine osteoarthritis change: a 6-year longitudinal study. Health Psychol 2010; 29:56.

- 46 Petrie KJ, Cameron LD, Ellis CJ et al. Changing illness perceptions after myocardial infarction: an early intervention randomized controlled trial. Psychosomat Med 2002; **64**:580–6.
- 47 Kirby JS, Butt M, Esmann S et al. Association of resilience with depression and health-related quality of life for patients with hidradenitis suppurativa. *JAMA Dermatol* 2017; **153**:1263–9.
- 48 Kirby JS, Sisic M, Tan J. Exploring coping strategies for patients with hidradenitis suppurativa. JAMA Dermatol 2016; 152:1166-7.
- 49 Zouboulis C, Tzellos T, Kyrgidis A. Development and validation of the International Hidradenitis Suppurativa Severity Score System (IHS4), a novel dynamic scoring system to assess HS severity. Br J Dermatol 2017; 177:1401–9.
- 50 Fotiou K, Hofmann M, Kaufmann R et al. Pictorial representation of illness and self measure (PRISM): an effective tool to assess the burden of psoriasis. J Eur Acad Dermatol Venereol 2015; **29**:2356–62.
- 51 Sensky T, Büchi S. PRISM, a novel visual metaphor measuring personally salient appraisals, attitudes and decision-making: qualitative evidence synthesis. PLoS ONE 2016; 11:e0156284.
- 52 Mühleisen B, Büchi S, Schmidhauser S et al. Pictorial Representation of Illness and Self Measure (PRISM): a novel visual instrument to

measure quality of life in dermatological inpatients. Arch Dermatol 2009; **145**:774-80.

53 Rosenthal R. Meta-Analytic Procedures for Social Research (Revised). Thousand Oaks, CA: Sage, 1991.

Supporting Information

Additional Supporting Information may be found in the online version of this article at the publisher's website:

Table S1 Summary of the hierarchical multiple regressionanalysis explaining depression scores.

Table S2 Summary of the hierarchical multiple regression

 analysis explaining anxiety scores.

Table S3 Summary of the hierarchical multiple regressionanalysis explaining impaired quality-of-life scores.

Powerpoint S1 Journal Club Slide Set.